Author’s response to reviews

Title: Integrated analysis of lncRNA-miRNA-mRNA ceRNA network in squamous cell carcinoma of tongue

Authors:

Rui-sheng Zhou (khaszrs@gzucm.edu.cn)
En-Xin Zhang (ergep53@126.com)
Qin-feng Sun (sunqinfeng825@126.com)
Zeng Jie Ye (zengjieye@qq.com)
Jian-wei Liu (kqyyljw@163.com)
Dai-Han Zhou (13902206571@139.com)
Ying Tang (18825144748@163.com)

Version: 4 Date: 16 Jul 2019

Author’s response to reviews:

Dear Editor,

First of all, we would like to extend our sincerest thanks to you and the reviewers for the insightful critiques and critical comments on our manuscript (Manuscript #:BCAN-D-19-00107R2) entitled “Integrated analysis of lncRNA-miRNA-mRNA ceRNA network in squamous cell carcinoma of tongue”. Your mentoring not only made great contribution for us to improve our work, but also considerably broadened our academic vision in our research area. Thus, our research group has already made a thorough revision exactly according to the comments and your suggestions.

Please also kindly inform me if you need further clarifications or supporting materials. We trust that our revised manuscript would satisfy the requirements of BioMed Research International. We sincerely wish that this manuscript would be accepted for publication. Thank you very much for your help and everything you have done for us. Hope we will hear from you soon. Thank you for considering this manuscript for publication in the journal.
Sincerely,

Ying Tang  
Institute of Tumor, Guangzhou University of Chinese Medicine, Guangzhou 510006, China;  
E-mail: tangying@gzucm.edu.cn;18825144748@163.com